Author’s response to reviews

Title: Short-term effectiveness of low dose liraglutide in combination with metformin versus high dose liraglutide alone in treatment of obese PCOS: randomized trial

Authors:

Mojca Jensterle (mojcajensterle@yahoo.com)
Nika Kravos (nikakravos@gmail.com)
Katja Goricar (katja.goricar@mf.uni-lj.si)
Andrej Janez (andrej.janez@kclj.si)

Version: 2 Date: 11 Jan 2017

Author’s response to reviews:

Prof. Andrej Janez, MD, PhD

Department of Endocrinology, Diabetes and Metabolic Diseases
University Medical Centre Ljubljana
Zaloška 7, Ljubljana 1525, Slovenia
Fax: 0038615229359
E-mail: andrej.janez@kclj.si

Ljubljana, 11th Januar 2017

Ref. Short-term effectiveness of low dose liraglutide in combination with metformin versus high dose liraglutide alone in treatment of obese PCOS: randomized trial (BEND-D-16-00195)

Mojca Jensterle, Nika Aleksandra Kravos, Katja Goričar, Andrej Janez
Dear Professor Michael O’Reilly,

We would like to sincerely thank you and the Reviewers for the valuable comments. We have substantially rewrite the whole Introduction Section, the first two paragraphs of Methods section, the report of Adverse events and the whole Discussion and Ethics approval and consent to participate Sections. As you commented an overlap in the Methods section is inevitable when describing similar methodology to that in our previously published study. In addition, the full name of the Ethics committee that approved our study was added and the date of registration and "Retrospectively registered" was included at Trial registration Section.

We believe that the revised manuscript fits well within the scope of BMC Endocrine Disorders and we hope that it will be accepted for publication.

The revised article is attached.

We look forward to hearing from you.

Sincerely yours,

Andrej Janez